

### PEER REVIEW HISTORY

*Gynecology and Obstetrics Clinical Medicine* publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

### ARTICLE DETAILS

<b>TITLE (PROVISIONAL)</b>	For cervical cancer screening, which test is better, and for whom?
<b>AUTHORS</b>	Mayeaux, E.J., Zhao, Yun